Author's response to reviews

Title: Conceptualization of Category-Oriented Likelihood Ratio: A Useful Tool for Clinical Diagnostic Reasoning

Authors:

Hamideh Moosapour (dr_moosapour@yahoo.com)
Mohsin Raza (Mohsinreza60@yahoo.com)
Mehdi Rambod (mehdi.rambod@gmail.com)
Akbar Soltani (soltania@tums.ac.ir)

Version: 15 Date: 2 October 2011

Author's response to reviews:

Date: 10/2/2011

MS: 1152643293376983 - Conceptualization of Category-Oriented Likelihood Ratio: A Useful Tool for Clinical Diagnostic Reasoning

Dear Prof. Aldcroft

We revised the manuscript and change the abstract structure according to your instructions for authors.

Respectfully yours,

Akbar Soltani

Associate Professor of Internal Medicine and Endocrinology
Tehran University of Medical Sciences
Evidence-Based Research Center
Email: soltania@tums.ac.ir